Author’s response to reviews

Title: The association between Colombian medical students’ healthy personal habits and a positive attitude toward preventive counseling: cross-sectional analyses

Authors:

John Duperly (jduperly@uniandes.com.co)
Felipe Lobelo (lobelo@mailbox.sc.edu)
Carolina Segura (seguracarolina@gmail.com)
Francisco Sarmiento (franciscosarmiento@gmail.com)
Deisy Herrera (deisymilenaherrera@gmail.com)
Olga L Sarmiento (osarmien@uniandes.edu.co)
Erica Frank (efrank@emory.edu)

Version: 4 Date: 18 June 2009

Author’s response to reviews:

Dear Mr Todd:

Enclosed you will find an editorial revision of our manuscript, “The association between Colombian medical students’ healthy personal habits and a positive attitude toward preventive counseling: cross-sectional analyses”. We have corrected the minor editorial concerns raised and have ensured that the manuscript conforms to all of the formatting checklist points.

Thank you very much

The authors